Part I Introduction

Introduction to Part I

The Italian Renaissance is an interesting and complex moment in the history of politics, society and culture. This volume contains thirteen essays designed to introduce readers to some of the most fundamental aspects of the period in question. I have organized these readings thematically because I still agree with the Swiss German historian Jacob Burckhardt, whose The Civilization of the Renaissance in Italy (1860) remains the classic study of this subject, that the Renaissance was not an event that unfolded in a series of neat chronological steps but a subject that emerged episodically. Part I offers an introductory overview to the Italian Renaissance and extensive, though by no means comprehensive, bibliography of some of the important work on this period in English. Part II examines politics and the Renaissance citystate. Part III introduces readers to crucial aspects of behavior in the cities. the physical location in which many activities that define the Renaissance occurred. Part IV examines the nature of marriage and motherhood and attitudes towards sexuality. Part V discusses the intellectual life of Renaissance Italy, offering three different perspectives on humanism. Finally, Part VI turns to the problem of patronage in relationship to the production of art and culture in the fifteenth and early sixteenth centuries.

These different themes by no means exhaust the variety of subjects that an introductory volume on Renaissance Italy might cover. But they are all essential for a better understanding of why this society continues to fascinate us. Each of the essays in this volume demonstrates some of the best work in political, social, and cultural history done in the past few decades. I have supplemented them with introductory essays, designed to raise general questions about the theme as a whole, and with images that reflect directly the themes of each section. Looking at art is a fundamental aspect of understanding Italian Renaissance history, not simply for the pleasure it affords but because it offers us crucial pieces of evidence about how this society envisioned itself.

Historical inquiry should be a process that yields multiple interpretations of a single subject. This volume is not designed to describe Renaissance Italy in any definitive sense, because there are many aspects of this world that a single book cannot cover in adequate detail. Rather its goal is to inform readers about key attributes of this world and to invite you to participate in the process of understanding and imagining this particular past. Since the modern idea of history – understanding the past in its own terms – was, in part, a creation of Renaissance humanism, this makes it all the more fitting that historians continue to exercise their skills on grappling with the problems and pleasures of a society that truly appreciated the value of a historical perspective.

Understanding the Italian Renaissance

Paula Findlen

No enterprise, no matter how small, can begin or end without these three things: power, knowledge, and love. Fourteenth-century Tuscan saying

I How Petrarch Rediscovered Italy

In 1337, the son of a Florentine merchant who had spent most of his life in the papal city of Avignon in southern France, made his first trip to Rome. Gazing upon the city that had once been the center of the Roman Empire and the heart of Christendom until the papacy moved to Avignon in 1309, Francesco Petrarca (1304–74) lamented the sorry condition of "the remains of a broken city." Already in his writings he anticipated a moment when future generations would emerge from "this slumber of forgetfulness into the pure radiance of the past."²

Petrarch's profound sense of displacement from his own times and his fierce desire to recapture the glories of a neglected past lay at the heart of the cultural movement we know as the Renaissance. He belonged to the earliest generations of humanists who had begun to re-examine the ancient

- 1 Vittore Branca, ed., Merchant Writers of the Italian Renaissance, trans. Murtha Baca (New York: Marsilio Publishers, 1999), p. ix. Readers wishing to consult other surveys of Renaissance Italy might start with Lauro Martines, Power and Imagination: City-States in Renaissance Italy (New York: Vintage, 1980, 1979), Peter Burke, The Italian Renaissance: Culture and Society in Italy (Princeton: Princeton University Press, 1986; 1972); and Lisa Jardine, Wordly Goods: A New History of the Renaissance (London: Macmillan, 1996). The most fundamental account remains Jacob Burckhardt, The Civilization of the Renaissance in Italy, trans. S. G. C. Middlemore (London: Penguin, 1990).
- 2 Petrarch, *Rerum familiarium libri I–VIII*, p. 294 (*Fam.* VI, 2): "Et euntibus per menia fracte urbis et illic sedentibus, ruinarum fragmenta sub oculis erant." In Petrarca, *Opere* (Florence: Sansoni, 1975), p. 484; and Petrarch, *Africa* (1336), in John T. Paoletti and Gary M. Radke, *Art in Renaissance Italy* (London: Laurence King, 1997), p. 26.

past in the study of law and literature and, in Petrarch's case, use the lessons he learned from antiquity to examine the spiritual malaise of the midfourteenth century.³ By the end of his life, he surrounded himself with the artifacts that evoked the world he wished to inhabit: manuscripts written by ancient Romans such as his beloved Cicero, to whom Petrarch composed imaginary letters in an effort to converse with the dead; ancient coins and medals; books of modern authors such as Dante Alighieri (1265–1322) and Giovanni Boccaccio (1313–75) who transformed Tuscan prose into a celebrated literary style; a painting by Giotto that was among his dearest possessions; and Greek manuscripts that he found himself unable to read because there was no one fully capable of unlocking their ancient grammar.

A Tuscan who spent most of his life outside of Italy, Petrarch exhibited a passionate craving for the best things that his society could offer him, ancient as well as modern. He attempted to perfect his Latin in imitation of Cicero in his letters, dialogues, and treatises, and celebrated the resurgence of his native Tuscan by writing poetry in this vernacular. He wrote dialogues between himself and Augustine, his favorite Christian writer, in order to explore the problems of a moral existence in a world rife with temptations, among them, the pleasure of knowledge itself. After numerous trips to Italy during which he collected manuscripts and met admirers such as Boccaccio, Petrarch finally returned to his native land. He initially settled in Milan in 1353, later moving to the small town of Arquà, south of Padua, in 1370 to enjoy his final years in proximity to one of the great centers of learning.

In retrospect, it might seem obvious to us that such cities as Florence, Venice, and Rome should be the centerpiece of the Italian Renaissance, because their leading historians and artists engaged so actively in their myth-making in the fifteenth and sixteenth centuries, but this was not immediately evident to contemporary observers. The Italy to which Petrarch returned was not Tuscany, but Milan, center of a thriving court culture, Verona, home of rich monastic libraries, Padua, a great center for law and scholarship, and of course Rome, a ghost of the past. His friend Boccaccio, another Tuscan who spent years away from Florence working for his family's business in Naples, felt a similar ambivalence towards Florence, though he returned there in 1341. Angevin Naples, he wrote with great nostalgia, had been "happy, peaceful, abounding in good things, magnificent, and ruled by a single king." Instead he found Florence filled with "as many opinions as there are men, and always under arms, and at war as much at home as abroad." Boccaccio eventually retreated to his

³ Ronald G. Witt, In the Footsteps of the Ancients: The Origins of Humanism from Lovato to Bruni (Leiden: Brill, 2000); and Charles Trinkhaus, The Poet as Philosopher: Petrarch and the Formation of Renaissance Consciousness (New Haven: Yale University Press, 1979).

birthplace of Certaldo in 1361, where he celebrated "the absence of the ambitions and the unpleasantness and annoyances of our town-dwellers." ⁴ He had no special love for fourteenth-century Florence nor any particular allegiance to its politics.

Following his death in 1374, Petrarch's home and tomb in Arquà became pilgrimage sites for his admirers who saw him as the embodiment of the cultural movement that another Tuscan, the painter and historian Giorgio Vasari (1511–74), would describe in 1550 as a *rinascita* – a rebirth. Boccaccio, too, would be celebrated as the embodiment of all that was great about Florence. However, Vasari did not have either writer in mind when he wrote these words. He referred specifically to the transformation of the arts between the fourteenth and the sixteenth centuries, beginning with Cimabue (ca. 1240–1302) and Giotto (1266/67–1337) and culminating in Michelangelo Buonarotti (1475–1564) who was still alive when Vasari completed his first edition. Sixteenth-century French humanists translated rinascita into renaissance to describe the general revival of learning and culture that they had witnessed. As French culture extended its influence throughout Europe in the seventeenth and eighteenth centuries, this foreign word stuck. In 1764 no less a figure than the great English historian Edward Gibbon described himself as seeing the "renaissance" of art as he gazed at the paintings in the Uffizi galleries and contemplated writing a history of the Medici. By the mid-nineteenth century historians such as Jules Michelet and Jacob Burckhardt used this concept to define the period as a whole.⁵ Burckhardt, in particular, insisted that it was not simply the revival of antiquity that constituted the Renaissance. If Petrarch wished to think and write like a Roman, it was a reflection of a much more fundamental transformation of his society.

In *The Civilization of the Renaissance in Italy* (1860), Burckhardt famously remarked: "The Renaissance would not have been the process of world-wide significance which it is, if its elements could be so easily separated from one

⁴ Giovanni Boccaccio, *Fiametta*, as quoted in Vittore Branca, *Boccaccio: The Man and His Works*, trans. Richard Monges and Dennis J. McAulifee (New York: New York University Press, 1976), p. 58; idem, *Epistola consolatoria a Pino de' Rossi*, in Branca, *Boccaccio*, p. 129. For a fine account of why Naples was appealing, see Jerry H. Bentley, *Politics and Culture in Renaissance Naples* (Princeton: Princeton University Press, 1987).

⁵ Vasari alluded to the "progress of art's rebirth" in the preface to his *Lives of the Most Eminent Italian Architects. Painters, and Sculptors from Cimabue to Our Times,* first published in 1550 and reissued with important modifications in 1568. See Giorgio Vasari, *The Lives of the Artists,* ed. and trans. Julia Conaway Bondanella and Peter Bondanella (Oxford: Oxford University Press, 1991), p. 6. On the idea of the Renaissance, see especially Wallace K. Ferguson, *The Renaissance in Historical Thought: Five Centuries of Interpretation* (New York: Houghton Mifflin, 1948); and Paula Findlen, *A Fragmentary Past: The Making of Museums and the Making of the Renaissance* (forthcoming).

another." 6 While we may no longer agree with Burckhardt on the precise meaning of the Italian Renaissance, which he defined as the birth of the modern western world and modern man, his idea of the interconnectedness of Italian Renaissance society still stands. If today most of us primarily experience the Italian Renaissance through viewing its great works of art in museums and in countless reproductions of paintings such as Sandro Botticelli's Birth of Venus (ca. 1485) and Leonardo da Vinci's Mona Lisa (1503–6), and by reading a handful of its most important books such as Boccaccio's Decameron (ca. 1350) and Machiavelli's The Prince (1513), we need to remember that these things are not timeless and isolated artifacts but were products of a specific historical moment. Understanding the Italian Renaissance allows us to appreciate how the cultural traces of the Renaissance that have survived emerged from its social, political, economic, and religious fabric. This context is well worth studying because Italy played a fundamental role in shaping the politics and culture of western Europe from antiquity through the seventeenth century.

Just as Petrarch's vision of the past is part of a more complicated story about the growth of humanist learning, which looked to ancient Greece and Rome for models of good politics and culture, the emergence of Florence, Venice, and Rome as Renaissance cities filled with great artists and generous patrons cannot be told in isolation from developments throughout the Italian peninsula in this crucial period of its history. No single individual or city gave birth to the Italian Renaissance or fully contained its achievements. Instead we might think of the Italian Renaissance as a Shakespearean drama in which the minor characters and the settings themselves were often the hinge of the entire story. In the past two decades, social and cultural history has given us a richer portrait of Renaissance Italy than we previously enjoyed. Many of the articles in this volume reflect this new understanding of the Renaissance. We now have a much better comprehension of how such issues as family and sexuality, friendship and patronage, knowledge and faith defined the parameters of behavior in this world in relationship to our continued appreciation of the great politics, art, and literature of the time. The chapters in this volume invite readers to explore the dynamic among these different dimensions of the Italian Renaissance.

2 The Italy that Petrarch Saw

But let us return to the beginning of the Italian Renaissance – the moment of Petrarch's great nostalgia. The world that Petrarch inhabited was a study in contrasts, and this was surely part of his disaffection from it. The poverty

of Rome, the only major city of this period whose walls were not constantly being rebuilt to accommodate a population bursting at the seams, appeared all the more incongruous in a landscape defined by urban growth and prosperity. Florence, already a city of over 100,000 inhabitants before the great plague pandemic of 1348 killed somewhere between one-third to one-half its population, typified this other face of Italy that Petrarch certainly knew well. It was a thriving commercial society that expressed its pride through great civic commissions such as the building of the Duomo, initiated in 1296 and halfway towards completion when Petrarch visited Florence, and the decoration of the south Baptistery doors that Andrea Pisano completed in 1336. To the south of Florence, the government of Siena commissioned Ambrogio Lorenzetti to paint his Allegory of Good and Bad Government (1338–40), a subtle account of the many forces that shaped a city, on the walls of the Palazzo Pubblico (Plate 1). To the north, noble families such as the d'Este in Ferrara, the Gonzaga in Mantua, and the Visconti in Milan had just begun to establish their rule over their respective cities. In Venice, artisans laid the foundation stones of the Doge's Palace in 1340. The growth of many Italian towns in the face of the papacy's abandonment of Italy, and despite the intermittent struggles between ruling families who used their allegiance to the pope or the emperor as a pretext for attacking their enemies, was one of the many paradoxes that troubled witnesses to Italy's fourteenth century.

In contrast to the nation-states that emerged in Spain, France, and England during the Renaissance, Italy would not become a nation until 1860, the year Burckhardt completed his great study of Renaissance Italy. and this new Italy did not include the Papal States until 1869. Although Renaissance Italy lacked any prospect for unified political leadership and seemed to have temporarily ceded its religious primacy to the French, it surpassed all other parts of Europe, with the possible exception of Flanders, in its commercial economy. "Money is the vital heat of a city," observed the Franciscan preacher Bernardino da Siena (1380–1444). A steady stream of transactions involving florins, ducats, and bills of exchange, all neatly recorded in account books that survive from this period, defined the world in which the majority of city-dwellers lived. Boccaccio vividly captured this world in his tales of greedy and gullible merchants who betrayed and were betrayed by their wives while engaging in commerce far from home. He had direct experience of such things working as a banker's apprentice in Naples. "Make me this calculation," a fourteenth-century Venetian merchant wrote repeatedly in his notebook. The connections among cities – linked by trade, immigration, and politics – made Italian society thrive.

⁷ John E. Dotson, ed. and trans., *Merchant Culture in Fourteenth Century Venice: The Zibaldone da Canal* (Binghamton, NY: Medieval and Renaissance Texts and Studies, 1994), p. 29 and *passim*.



Plate I Ambrogio Lorenzetti, *The Good City-Republic* (1338–40). *Source*: Sala dei Nove, Palazzo Pubblico, Siena (partial view of east wall).

This panel is part of Ambrogio Lorenzetti's (ca. 1295–1348) famous fresco cycle, most commonly known as the Allegory of Good and Bad Government. Commissioned by the Nine who ruled Siena from 1287 to 1355, it offers an idealized portrait of the virtues of republican government. Lorenzetti's fresco transforms details of an Italian city-state into a utopian image of a harmonious and prosperous society. The fact that Lorenzetti's fresco cycle was located in the Room of the Nine in the Palazzo Pubblico, the seat of government in Siena, suggests its importance as a work of public art. City officials and visitors who entered the room were first confronted with an image of a violent and chaotic city ruled by Fear, "The City-State Under Tyranny" (west wall). Looking at the north wall, they saw "The Virtues of Good Government," a fresco depicting how communal government dispensed justice, maintained peace, and preserved the common good. Finally they turned to the image of "The Good City-Republic," where Security maintains peace by guarding the city gates that open into the countryside. Inside the walls, the social and economic life of the city thrives. Note the depiction of Siena's Duomo in the upper-left corner as well as the presence of aristocratic family towers that suggest the kind of oligarchical power that controlled the city. Building projects, a lively market, and a marriage procession (lowerleft corner) offer a snapshot of an Italian city on the eve of the Black Death.

The merchant community in Avignon, for example, was filled with all manner of Italians, trading goods, opening branches of their banks, and offering services to an international clientele. When Francesco Datini (ca. 1335–1410) decided to become a merchant, he opened his first shop in Avignon, followed by branches in Pisa, Genoa, Valencia, and Majorca. But he also made sure to become a Florentine citizen after returning to Italy in 1382, and eventually left all of his wealth to his native city of Prato upon his death. Like Petrarch, he could not spend his entire life outside Italy without eventually seeking recognition at home.

More often than not, citizenship in the Italian Renaissance cities was defined by long absences for its male members, who stayed away from home so that their family businesses might prosper and, at times, because the vicissitudes of politics had forced them into exile. In July 1449, for instance, Alessandra Strozzi lamented the fact that her youngest son, thirteen-year-old Matteo, was about to depart for Naples to learn his trade, following his two elder brothers who also had left Florence in pursuit of honor and profit. The mobility of Italian merchants and patricians brought the values and practices of the Italian cities into contact with many parts of the world. For all these reasons, Petrarch could experience Italy even from the distance of Avignon, because so much of Italy had migrated to southern France with the papacy's abandonment of Rome.

The commercial economy of northern and central Italy produced forms of governance that reflected the emergence of a wealthy urban elite and a society in need of revenue to fund the costs of intermittent warfare. Communal government shaped the politics of many Italian city-states as early as the twelfth century, though there were other cities ruled by hereditary princes with feudal land holdings. The cities of Tuscany, principally Florence, Siena, and Pisa, were ruled by an elected citizen elite who levied public taxes to support the enterprise of the state. Machiavelli would later observe with pride that Tuscany contained three republics and "no baronial castles." Their merchants and bankers traveled all over the Mediterranean, Levant, and northern Europe. The maritime republics of Genoa and Venice, both as large as Florence, rested their prosperity on trading empires

⁸ Bernardino da Siena's sermons are quoted in Robert Bonfil, *Jewish Life in Renaissance Italy*, trans. Anthony Oldcorn (Berkeley: University of California Press, 1994), p. 24. On Datini, see Iris Origo, *The Merchant of Prato: Daily Life in a Medieval City* (Harmondsworth: Penguin, 1986; 1957).

⁹ Alessandra Strozzi, Selected Letters of Alessandra Strozzi, ed. and trans. Heather Gregory (Berkeley: University of California Press, 1997), p. 45.

¹⁰ Niccolò Machiavelli, *The Discourses*, ed. Bernard Crick, trans. Leslie J. Walker, S. J. and Brian Richardson (London: Penguin, 1983; 1970), p. 243. This work seems to have been written between 1516 and 1519, though it was already in gestation when Machiavelli completed *The Prince*.

that competed with each other for control of transport and commerce between East and West. *Januensis ergo mercator*, "Genoese therefore a merchant," went a saying of the time. Venice's fifteenth-century chroniclers noted with pride that Venetian patricians "continue to trade." They praised their elected head of government, the doge, for having the appearance of a prince, the virtues of a citizen, and the temperament of a merchant.¹¹

City-states increasingly were territorial states whose geography reflected the ambitions of their leading citizens. In Petrarch's time, the expansion of various states had just begun and would be completed in the fifteenth and early sixteenth centuries. The Florentines, for example, annexed Pisa in 1406 and Livorno in 1421 with the specific goal of acquiring access to the sea, as the culmination of a territorial state that had begun with the submission of Prato to Florentine rule in 1350. The Venetians declared their primacy over the Genoese by establishing an overseas empire in the Eastern Mediterranean that included much of Istria, Dalmatia, Crete, and colonies scattered between Venice and Constantinople. The size of their territorial state within Italy, beginning with the capture of Treviso in 1389 and culminating in the acquisition of Bergamo by 1428, deeply alarmed neighboring states such as Milan and Ferrara (see Map 1). Pope Pius II would write of the Venetians in 1459 that "they aim at the dominion of Italy and all but dare to aspire to the mastery of the world." But he was quick to add that if the Florentine state grew as strong, "they would also have an equal ambition for empire." 12 To a certain extent, the logic of the city-state reflected the dynamics of trade in which the wealthiest cities and rulers had the resources to secure their position within a region. Subject cities might occasionally rebel when they sensed weakness at the center, or when external threats from other rulers forced them to reconsider their allegiances, but they came to accept the idea that Italy would be a land of many would-be princes but few actual rulers.¹³

At the beginning of the Renaissance, Italy was a crossroads between northern Europe and the Levant. ¹⁴ Tuscan gradually emerged as a com-

- 11 Steven Epstein, *Genoa and the Genoese*, 958–1528 (Chapel Hill: University of North Carolina Press, 1996), p. xvi; and Marin Sanudo, *Praise of the City of Venice* (1493), in *Venice: A Documentary History* 1450–1630, ed. David Chambers and Brian Pullan (Oxford: Blackwell, 1992), p. 11. It was said of the Venetian doge: "He has the bearing of a prince, but in the Senate he is a senator, and in the marketplace a citizen." In Edward Muir, *Civic Ritual in Renaissance Venice* (Princeton: Princeton University Press, 1981).
- 12 Aenius Sylvius Piccolomini, *Memoirs of a Renaissance Pope: The Commentaries of Pius II (An Abridgment)*, ed. Leona C. Gabel, trans. Florence A. Gragg (New York: Capricorn Books, 1962; 1959), p. 352.
- 13 Two exemplary studies of subject cities are Judith C. Brown, *In the Shadow of Florence: Provincial Society in Renaissance Pescia* (Oxford: Oxford University Press, 1982); and James S. Grubb, *Firstborn of Venice: Vicenza in the Early Renaissance State*. Baltimore: Johns Hopkins University Press, 1988).
- 14 The classic study of this subject remains Fernand Braudel, The Mediterranean and the



Map I Renaissance Italy.

mercial language across the Mediterranean before becoming the literary language of educated Italians in the sixteenth century. Genoese and Venetian dialects helped to create the international language of seafarers. Societies in proximity to Italy also shaped its culture. In parts of northern Italy, French and German were the lingua franca, just as Arabic and Greek continued to be spoken in parts of southern Italy – vestiges of centuries of foreign rule and a reflection of the constantly shifting boundaries between Italy, France, and the Holy Roman Empire. The economic opportunities in Italy also attracted foreigners. When the French diplomat Philippe de Commynes visited Venice in 1495, he alluded to the prominence of Greek, German, and Turkish colonies in the city, when he remarked that "most of the people are foreigners." Slaves, mostly of Slavic and eastern European origin, were vet another foreign population who entered the households of wealthy patricians as servants – the ultimate example of how a commercial society connected different peoples.¹⁶ From this perspective, Italy seemed utterly lacking in a single unifying identity – an entrepôt more than a state - save for a lingering sense that all parts of the Italian peninsula had inherited a legacy and a language from the Roman empire and had vested interests that loosely bound them together.

The emphasis on local interests shaped Italy as a region of multiple identities, roughly unified by language, customs, and culture. If the idea of communal government distinguished Italy from the rest of Europe, it was by no means uniform or evident in every part of Italy. The Florentines, Venetians, and Genoese, for example, all considered themselves citizens of great republics. But the nature of republican rule expressed itself differently in each instance. While all cities restricted active citizenship to the population of adult males who paid taxes, some cities such as Florence made membership in a guild (organizations which regulated entry into the major professions within a city) a crucial factor. Other cities made the question of lineage a primary criterion for political participation. The most famous case is Venice, whose "Golden Book" (*Libro d'oro*) recorded the names of all descendants of the city's ruling families since 1297, defining for the next five hundred years who could hold office. In 1377 the Visconti rulers of Milan created a similar book to distinguish their nobility, though it did not

Mediterranean World in the Age of Philip II, trans. Siân Reynolds (New York: Harper and Row, 1972–3), 2 vols. See also David Abulafia, The Two Italies: Economic Relations between the Norman Kingdom of Sicily and the Northern Communes (Cambridge: Cambridge University Press, 1977).

15 John Martin and Dennis Romano, ed., Venice Reconsidered: The History and Civilization of an Italian City-State 1297–1797 (Baltimore: Johns Hopkins University Press, 2000), p. 20.

16 This important subject is only just beginning to receive more attention. See Iris Origo, "The Domestic Enemy: Eastern Slaves in Tuscany in the Fourteenth and Fifteenth Centuries," Speculum 30 (1955): 321-66; and Robert C. Davis, "Slave Redemption in Venice, 1585–1797," in Martin and Romano, Venice Reconsidered, pp. 454–87.

have the same sort of lasting impact on the city's politics. ¹⁷ Both in the northern and southern-most parts of Italy, lineage mattered in ways that seemed less apparent in Tuscany.

The Florentines were suspicious of their city's nobility. They excluded magnates from holding office in 1293, out of fear that they would abuse their power. Their towers, that defined the political divisions of the medieval city, were slowly being toppled to make way for new civic buildings and, with them, a new style of politics. 18 "I am not a prince," Cosimo de' Medici (1389–1464) explained to pope Pius II (1458–64), carefully underscoring the distinction between the basis of his unofficial control of Florentine politics and hereditary rule. Cosimo also distinguished the legitimacy of his position, as unofficial ruler of Florence (1434–69), from rule by force. "A republic cannot be run in the same way as a despotic regime," he told Francesco Sforza, who had wrested the duchy of Milan from the hands of the Visconti in 1450. 19 He was acutely sensitive to the political differences within the Italian city-states and to the importance of communal traditions in his own city, as were the Venetians who declared that because they were a republic "everything must be done by ballot." The Venetian doge, elected as head of state for life in a process as complex and embattled as the election of a new pope by a cardinals' conclave, was decidedly not a prince who could act without the consent of the Great Council.²¹ Rather, he was their first citizen, a prince when the republic needed a princely figurehead but always a citizen constrained by the rules of citizenship in a Renaissance republic.

- 17 The best starting point for understanding Venetian history is still Frederic C. Lane, Venice: A Maritime Republic (Baltimore: Johns Hopkins University Press, 1973). For a recent reassessment, see Martin and Romano, Venice Reconsidered; and Gary Wills, Venice, Lion City: The Religion of Empire (New York: Simon & Schuster, 2001). The Venetian and Milanese registers are discussed in Stanley Chojnacki, Women and Men in Renaissance Venice: Twelve Essays on Patrician Society (Baltimore: Johns Hopkins University Press, 2000), p. 30 and passim. The most accessible study of Milanese court culture is Gregory Lubkin, A Renaissance Court: Milan under Galeazzo Maria Sforza (Berkeley: University of California Press, 1994).
- 18 Gene Brucker, *Renaissance Florence* (Berkeley: University of California Press, 1983; 1969), p. 133; and Carol Lansing, *The Florentine Magnates: Lineage and Faction in a Medieval Commune* (Princeton: Princeton University Press, 1991). Around 1200, Florence had more than 150 towers, and even today cities such as Bologna and San Gimignano contain a few of the aristocratic family towers. "Magnate" is a term for the older ruling families of Florence, in contrast to newer families such as the Medici who came to rule the city.
- 19 Dale Kent, Cosimo de' Medici and the Florentine Renaissance: The Patron's Oeuvre (New Haven: Yale University Press, 2000), p. 348.
- 20 Lunardo Emo to doge Andrea Gritti, ca. 1528–29, in Martin and Romano, Venice Reconsidered, p. 170.
- 21 The Great Council of Venice consisted of all the adult males of patrician families (approximately 3,000). The real decision-making bodies were the Senate and the Council of Ten. See Robert Finlay, *Politics in Renaissance Venice* (New Brunswick: Rutgers University Press, 1980).

It is a mistake to think of the Italian republics and their claims to reinvent Roman traditions of liberty as the dominant approach to Renaissance politics, when it was one of several viable alternatives. Numerous other Italian states were ruled by dukes, marquises, and aristocratic lords who built fortresses and castles to dominate the lands they held. Giovanni Antonio di Blazo Orsini, prince of Otranto, duke of Bari, and count of Lecce and Conversano reputedly commanded four hundred castles, according to one chronicler in 1444, and could ride from Naples to Taranto entirely on his own lands.²² Other states thrived under the leadership of mercenary soldiers (condottieri), who commanded powerful armies and maintained strategic allegiances with neighboring city-states. The Sforza in Milan, the Montefeltro in Urbino, the Malatesta in Rimini, Cesare Borgia in Romagna, and many lesser-known princes of this period were cut from this cloth. Such changes led pope Pius II to declare famously that in Italy "a servant can easily become a king."23 As a Sienese aristocrat, Pius II disliked the ease with which merchants and soldiers could become heads of state. And vet Machiavelli would complain what a socially closed world a city like Florence could be for the son of a poor country gentleman, rendering it difficult for a man without status, money, and connections to get a decent post in government. So we must not take Pius's assessment to mean that anyone could aspire to rule.

Italy, in other words, was a region of many different kinds of governments, elective and hereditary, communal and despotic, secular and religious, local and foreign. At the beginning of the thirteenth century there were over two hundred city-states in the Italian peninsula, perhaps as many as three hundred. The majority of them were in northern and central Italy, where approximately one-third of the population lived in towns or cities. Southern Italy was more rural, often ruled by foreign monarchs (both French and Spanish), and dominated by aristocratic, land-holding families whose castles still mark the terrain of such regions as Lazio, Campania, Calabria, and Puglia. The number of city-states was considerably diminished

²² Denys Hay and John Law, *Italy in the Age of the Renaissance 1380–1530* (New York: Longman, 1989), pp. 186–7.

²³ Burckhardt, Civilization, p. 33.

²⁴ Burke, *Italian Renaissance*, p. 209; and Benjamin G. Kohl and Alison Andrews Smith, *Major Problems in the History of the Italian Renaissance* (Lexington, MA: D.C. Heath, 1995), p. 56. See Philip Jones, *The Italian City-State: From Commune to Signoria* (Oxford: Clarendon Press, 1997).

²⁵ The feudal aspects of Renaissance Italy were not confined to the Kingdom of Naples and Sicily, however, but could also be found in parts of northern Italy such as the regions surrounding cities such as Ferrara and Mantua, and the Friuli region north of Venice. See Trevor Dean, Land and Power in Late Medieval Ferrara: The Rule of the Este, 1350–1450 (Cambridge: Cambridge University Press, 1988); Edward Muir, Mad Blood Stirring: Vendetta and Factions in

by Petrarch's time and would decline further in the next two centuries, as powerful rulers and cities absorbed smaller communes into their territorial states. To take one example, Siena, the great medieval banking center, had become a shadow of its former self by the fifteenth century, a small town of about fifteen thousand depopulated by plague, besieged by mercenary companies, and coveted by ambitious rulers who hoped to acquire it, as the Medici finally did in 1557.

Machiavelli would later complain that the problem with Italy was that everyone aspired to be a prince and, as result, each one impeded the other from ever realizing this goal. But he did not cease to dream of Italy as a cultural ideal that might be realized from its Roman inheritance. The last lines of The Prince came from a poem by Petrarch: "the ancient valor is not vet dead in Italian hearts."27 Almost two hundred years after Petrarch lamented the condition of Rome, Machiavelli continued to see Italy as a region whose antiquity might inspire a new modernity. He had experienced the cultural transformation of Italy that occurred in the fifteenth century as well as the political turbulence of the period between 1492 and 1527, when internal quarrels among Italian rulers and foreign invasions had thrown the Italian peninsula into chaos, once again. Yet he still believed in the greatness of Italy's cities and the innovations they produced, because he had lived in an age in which they had been the marvel of Europe. Florence, "the heart of Italy," Venice, "a terrestrial paradise," Rome, "the head of the world" (caput mundi).²⁸ This was the Italy that emerged in the century following Petrarch's return home.

3 Cities of Wood Become Marble

Even as the vitality of the late medieval cities gave way to the emergence of a smaller number of territorial states, dominated by the largest cities (Florence, Venice, Milan, Rome, and Naples), what remained was a thriving urban culture. The population of the Italian cities, though greatly dimin-

Friuli during the Renaissance (Baltimore: Johns Hopkins University Press, 1993); and Tommaso Astarita, The Continuity of Feudal Power: The Carracciolo di Brienza in Naples (Cambridge: Cambridge University Press, 1992).

- 26 William Caferro, Mercenary Companies and the Decline of Siena (Baltimore: Johns Hopkins University Press, 1998).
- 27 Niccolò Machiavelli, *The Prince*, trans. George Bull (London: Penguin, 1961), p. 138. The full stanza in Petrarch is: "Virtù contro a furore/Prenderà l'arme, e fia el combatter corto;/ Che l'antico valor/Nell'italici cor non è ancor morto."
- 28 Felix Gilbert, Machiavelli and Guicciardini: Politics and History in Sixteenth Century Florence (New York: Norton, 1984; 1965), p. 30; Finlay, Politics, p. 17; and Stinger, The Renaissance in Rome.

ished after the arrival of plague, gradually increased during the fifteenth and early sixteenth centuries to the levels prior to 1348. Such a demographically challenged age offered great scope for moralists who argued that marriage, as one fifteenth-century humanist put it, was "the fundamental union, which by its multiplication makes the city." For cities to thrive, families had to flourish. Public dowry funds (*Monte delle doti*), established in many cities in the fifteenth century, offered a good example of intersections between public and private interests. Citizens provided the government with income to spend on maintaining the state in war and peace. In return, a modest investment at the time of birth was supposed to yield a healthy dowry by the time a daughter was of marriageable age, though the expenditures of state often exceeded the promised returns (Plate 2). ³⁰

The size of the population not only influenced the marriage market and economic opportunities for urban residents but also played a role in determining the power of different groups. A city, after all, was a microcosm of human ambition. Nobles, patricians, scholars, clerics, artists, and artisans lived and worked cheek to jowl in the streets and piazzas, while the inhabitants of the surrounding countryside routinely traveled through the city gates, bringing the fruits of the countryside into the marketplace and exporting aspects of urban life into their own world. A large portion of the population did not enjoy the benefits of citizenship as part of living in the city. Yet each sector of Renaissance society found a means to turn the dynamic between patricians and popolani (ordinary city-dwellers), and the city and the countryside, to his or her own advantage. "Never trust anyone to the point where he can unmake you," wrote one merchant in a diary written for his heirs. "Try to stay close to whoever is in power." 31 Such advice about how to survive life in a Renaissance city indicates how the politics of the Italian city-states left urban inhabitants calculating to what degree their relationships with each other furthered their own interests.

However wary urban residents were of each other, they were not indifferent to the collective effect of their activities on the image of the city itself.

²⁹ David Herlihy and Christiane Klapisch-Zuber, *Tuscans and Their Families: A Study of the Florentine Catasto of 1427* (New Haven: Yale University Press, 1985), p. 230. This statement was made by Leonardo Bruni in 1436.

³⁰ Anthony Molho and Julius Kirshner, "The Dowry Fund and the Marriage Market in Quattrocento Florence," *Journal of Modern History* 50 (1978): 403–38.

³¹ Giovanni di Pagolo Moretti, *Ricordi* (early fifteenth century), in Branca, *Merchant Writers*, pp. 68, 71. The dynamics of urban behavior have been especially well explored in Richard Trexler, *Public Life in Renaissance Florence* (New York: Academic Press, 1980); and Ronald F. E. Weissman, *Ritual Brotherhood in Renaissance Florence* (New York: Academic Press, 1982); and Dennis Romano, *Patricians and Popolani: The Social Foundations of the Venetian Renaissance State* (Baltimore: Johns Hopkins University Press, 1987).



Italian Renaissance cities were celebrated throughout western Europe as marvels of architectural design and urban planning. During the fifteenth century, as the Florentine humanist and architect Leon Battista Alberti (1404–72) eloquently observed, cities of wood became cities of marble. If the quintessential building projects of the late Middle Ages had been churches, communal palaces, and merchant halls, Italian Renaissance building was defined by the urban palace. In his *Panegyric to the City of Florence* (1403–4), the humanist Leonardo Bruni (1370–1444), who became chancellor of the city in 1427, praised Florence's "wealth of build-

Plate 2 Vittore Carpaccio, Two Venetian Ladies on a Terrace (ca. 1495). Source: Museo Civico Correr, Venice.

Vittore Carpaccio (ca. 1465–1526) was known for his ability to capture the ordinary fabric of everyday life in his adopted city of Venice. His portrait of two Venetian women has sometimes been interpreted as a portrait of high-class courtesans in a city famous for its prostitutes. More likely, it is an image of the forced leisure of Venetian patrician women, increasingly confined to their homes to preserve their family's honor. Such an image brings to mind the crucial advice offered by the Venetian humanist Francesco Barbaro (1390–1454) in On Wifely Duties (1416): "I would have wives be seen in public with their husbands, but when their husbands are away wives should stay at home." The ambiguity of Carpaccio's painting also captures one of the fundamental debates in Renaissance society at this time: how to distinguish between class and wealth. The growth of sumptuary laws in many Italian cities reflected an effort on the part of the urban elite to maintain their place in the order of things by specifying the use of clothing, jewelry, and dowries according to social status, religion, and occupation. The zoccoli (platform shoes) the women are wearing made it virtually impossible for them to walk without being attended by servants.



ings," including "the homes of the private citizens."³² Merchant families, hereditary princes, and despots all contributed to the material transformation of various cities. Private family palaces and chapels became a measure of one's status in the city and often defined the territory that a family controlled. By the late fifteenth century Renaissance Italy was gripped with a building fever that transformed small towns as well as major cities. Duke Federico da Montefeltro (1420–82) took such a personal interest in designing the ducal palace that formed the nucleus of Urbino in the 1460s that many considered him to be its architect. Lorenzo de' Medici (1449–92) was sufficiently intrigued by reports of its splendor that he had an architect send him copies of the plans. Under duke Ercole I d'Este's rule in Ferrara (1471–1505), new palaces, streets, and fortresses transformed this small city into one of the greatest Renaissance courts in Europe.³³ The act of building expressed well the competitive spirit of the Italian city-states that looked to each other in establishing their reputations.

³² Leon Battista Alberti, *On the Art of Building* (written ca. 1450, published 1485), in Alison Cole, *Virtue and Magnificence: Art of the Italian Renaissance Courts* (New York: Harry N. Abrams, 1995), p. 21; and Leonardo Bruni, *Panegyric to the City of Florence* (1403–4), in *The Earthly Republic: Italian Humanists on Government and Society* ed. Benjamin G. Kohl and Ronald G. Witt with Elizabeth B. Welles (Philadelphia: University of Pennsylvania, 1978), pp. 139–40.

³³ Thomas Tuohy, *Herculean Ferrara: Erco d'Este (1471–1505) and the Invention of a Ducal Capital (Cambridge: Cambridge University Press, 1996).*

Wealthy citizens also expressed their civic pride and piety through sponsorship of monasteries, churches, hospitals, and other buildings that contributed to the public good. The Florentine patrician Giovanni Rucellai, an important patron of Alberti, felt that he had "given myself more honor, and my soul more satisfaction by having spent money than by having earned it." He was not alone in this belief. Adding up the staggering sums that his predecessors had spend on building, charity and taxes, a youthful Lorenzo de" Medici observed in 1472 that "although many people might think it would be better to have something in your purse, I think that it brings honor to our State, and I believe the money was well spent, and I am well content with this." Both understood that public munificence was a direct expression of economic capital and political power.

The effect of this new kind of urban patronage on many cities was quite visible to a new generation of pilgrims and tourists who came to Italy. Rome, a city where cows wandered through the muddy ruins of the Colosseum in the fourteenth century, emerged as a splendid center, thanks to the building projects of various popes from Nicholas V (1446-55) to Sixtus V (1585–90). Only after the papacy returned from Avignon in 1377, and survived the turbulent era of the Great Schism (1378–1417) when different factions recognized different popes, did Rome become the nucleus of the Papal States. In the same era when the Medici controlled Florence and the Sforza ruled Milan, the papacy extended its territorial claims to encompass much of central Italy. Rome emerged as an urban center to rival any of the other Italian cities, as Roman nobles and cardinals rushed to build palaces that incorporated pieces of Roman ruins into the design to cater to the new taste for the antique. By the early sixteenth century artists and humanists flocked to the Eternal City to associate themselves with the papal court under Julius II (1503–13) and Leo X (1513–21), just as they migrated to other cities such as Florence, Ferrara, Mantua, and Urbino where the ruling elite invested in culture.³⁵ If Renaissance Italy appeared in retrospect to be a cultural unity, it was because so many Italian rulers employed the same individuals at different moments in their careers, due to the reputations they had developed elsewhere.

The building culture of the Italian Renaissance cities demanded a wide array of specialized skills from its ordinary citizens whose labor realized

³⁴ Richard Goldthwaite, *The Building of Renaissance Florence: An Economic and Social History* (Baltimore: Johns Hopkins University Press, 1980), p. 85; Branca, *Merchant Writers*, p. 157 (March 15, 1472).

³⁵ Ingrid D. Rowland, *The Culture of the High Renaissance: Ancients and Moderns in Sixteenth-Century Rome* (Cambridge: Cambridge University Press, 1998); Charles L. Stinger, *The Renaissance in Rome* (Bloomington: Indiana University Press, 1985); and Peter Partner, *Renaissance Rome* 1500–1559: A Portrait of a Society (Berkeley: University of California Press, 1976).

dreams of great buildings filled with magnificent art.³⁶ This raises a fundamental point about our image of the Italian Renaissance: the books, paintings, sculptures, and buildings that give shape to the Italian Renaissance were all products of manual labor, as were the fine clothes, jewelry, wedding chests, mirrors, and furniture which adorned the bodies and homes of Renaissance Italians. Italian cities thrived because of the presence of skilled labor. The increased amount of legislation dealing with sumptuary laws, for instance, not only reflected anxieties about religion and social status – as expressed through attempts to regulate expenditures on clothes, jewelry, and dowries – but also underscored the essential fact that there was a wide variety of goods and services that one could acquire in a city and a population ready to expend their florins and ducats in acts of conspicuous consumption.

Knowledge in a Renaissance city also had its artisanal component. Prior to the advent of the printing press, which first appeared in Europe around 1450 and arrived in Italy in 1465, making books was a physical act that required skilled copyists. One of the most knowledgeable men in fifteenthcentury Italy was a Florentine bookseller, Vespasiano da Bisticci (1421–98), who thoroughly disdained the printed book as a poor substitute for a handmade copy. Famed throughout northern Italy for his work, he knew everyone in power because he provided the labor that made Renaissance libraries come into existence, just as the Venetian printer Aldus Manutius (ca. 1449–1515) would do for the next generation that sat in his printing house to assist him in the task of creating the ideal Renaissance library in print. When Cosimo de' Medici requested a library of 200 books, Vespasiano employed forty-five scribes for twenty-two months to fill this order.³⁷ The ability of booksellers to mediate between literate patrons and the artisanal world of book production gave them unique insights into the place of knowledge in Renaissance society. In this respect, Vespasiano was not unlike a younger Tuscan, Leonardo da Vinci (1452–1519), whose success with a number of his patrons rested on his ability to translate his skills into practical expertise in architecture, engineering, and armaments in addition to his abilities as a painter. Thus, on many levels, Renaissance Italy was a world of skilled artisans who worked in cities, served princes at court and on the battlefield, and found ways to enrich their families by making their particular expertise essential to their society.³⁸

³⁶ Goldthwaite, Building of Renaissance Florence.

³⁷ Brian Richardson, *Printing, Writers and Readers in Renaissance Italy* (Cambridge: Cambridge University Press, 1999), p. 3. By contrast, Alberti claimed in 1466 that the same number of books could be printed in one hundred days.

³⁸ On Italian artisans, see Samuel K. Cohn, Jr., *Laboring Classes in Renaissance Florence* (New York: Academic Press, 1980); and Robert C. Davis, *Shipbuilders of the Venetian Arsenal: Workers and Workplace in the Preindustrial City* (Baltimore: Johns Hopkins University Press, 1991).

Written words were among the most valuable commodities in Renaissance society. The inhabitants of the Italian cities were unusually literate because writing, like counting, was a practical skill. Approximately one-third of urban males and a smaller percentage of women were literate. ³⁹ One of the most important figures in an Italian city was the notary whose ability to read and write allowed him to memorialize a wide variety of ordinary human interactions concerning marriage, dowries, property, business, taxes, and inheritance. The written word was the third party to virtually every transaction of significance that occurred in an Italian city. Those who possessed the ability to read and write had access to an entirely different level of society.

Writing was not simply a skill of the elite but a manifestation of a society that documented all of its activities. The Sienese cloth-cutter Benedetto di Bartolomeo, who transcribed the forty-five sermons that Bernardino da Siena gave in his native city in the summer of 1427, understood well that his literacy was a special gift. He described himself as a man with "a wife and several children, few possessions, and much virtue."40 Benedetto might have appreciated the decision of a merchant's wife, Margherita Datini, to have the family notary. Ser Lapo Mazzei, teach her how to read and write in her thirties. Her husband Francesco worried that some of these skills were not entirely useful. "Do not pay so much heed to reading that you do all other things ill," he counseled. Datini did not equate literacy with the cultivation of learning but saw it as a necessary part of business. Reading, of the sort Petrarch had in mind when he talked about his desire to know Cicero, was not exactly the kind of literacy that Datini valued. He was not unlike wealthy peasants who, unable to read or write themselves, nonetheless understood the power of words when they asked others to record their important transactions for them.⁴¹

We should not assume, however, that the two kinds of literacy were mutually exclusive. Just as the busy commercial life of the city produced a new building aesthetic, it also encouraged new attitudes towards learning. One Florentine merchant recorded how he copied favorite passages from Boccaccio in his notebooks "during the plague epidemic of 1449 for my own pleasure." Copying Petrarch's vernacular poems was also a favorite

³⁹ Paul F. Grendler, Schooling in Renaissance Italy: Literacy and Learning 1300–1600 (Baltimore: Johns Hopkins University Press, 1989), p. 77.

⁴⁰ Franco Mormando, *The Preacher's Demons: Bernardino of Siena and the Social Underworld of Early Renaissance Italy* (Chicago: University of Chicago Press, 1999), p. 43.

⁴¹ Origo, *Merchant of Prato*, p. 213. For a fascinating discussion of peasant literacy, see Duccio Balestracci, *The Renaissance in the Fields: Family Memoirs of a Fifteenth-Century Peasant*, trans. Paolo Squatriti and Betsy Merideth (University Park, PA: Pennsylvania State University Press, 1999).

⁴² Kent, Cosimo de' Medici, p. 71.

pastime of those who could read and write. Literature, history, advice, and finance appeared together indiscriminately in the merchant diaries and account books of the Italian Renaissance.

The Florentine merchant Giovanni di Pagolo Morelli expressed his own sense of Petrarch's values when he noted that behaving like the Romans was generally not useful in everyday life, "especially in the things with which we have to deal, which are more material than the great events of Rome." But he made an exception for politics as an arena in which one should think and behave like a Roman. "For as we are descended from them in essence we should show this in virtue and substance as well."43 Such comments, inscribed in notebooks as lessons for sons to learn from their fathers, suggest the ways in which ordinary citizens were able to translate Petrarch's high-minded ideals into practical results. "Books ring with the voices of the wise," observed cardinal Bessarion in 1468 as he donated his great library to the city of Venice. "They are full of the lessons of history, full of life, law, and piety. The live, speak, and debate with us; they teach, advise, and comfort us; they reveal matters which are furthest from our memories, and set them, as it were, before our eyes."44 The learning of the schools helped to shape the discourse of public life, but reading a tale from Boccaccio could be the consolation of a merchant in an hour of despair.

In the same period in which cities of wood became marble, their citizens found a new voice in the sort of public eloquence that Petrarch hoped to inspire. Humanist learning was primarily the domain of educated patricians – lawyers, doctors, theologians, notaries, and scholars who made a living either teaching or working in the service of the state as secretaries – and princes who knew their Cicero, Livy, Virgil, Aristotle, and other ancient writers so well that they could cite them line and verse. They had studied with famous masters like Guarino of Verona (1374–1460), who had skills that Petrarch only dreamed of and provided his pupils with a thorough grounding in the studia humanitatis. This term, borrowed from Cicero, encompassed grammar, rhetoric, poetry, history, and moral philosophy. Guarino's reputation rested on his association with the famous Greek scholar Manuel Chrysoloras. He traveled with him to Constantinople before 1403 and 1408, returning laden with Greek manuscripts that promised to unlock further the secrets of antiquity. By the 1430s his school was flourishing in Ferrara, as was the school of Vittorino da Feltre (ca. 1378-ca. 1446) in Mantua and many others of this kind in different Italian cities. 45

⁴³ Giovanni di Pagolo Morelli, Ricordi, in Branca, Merchant Writers, p. 75.

⁴⁴ Chambers and Pullan, Venice, p. 357.

⁴⁵ Charles G. Nauert, *Humanism and the Culture of the Renaissance* (Cambridge: Cambridge University Press, 1995), pp. 35, 46–8. See also Anthony Grafton and Lisa Jardine, *From Humanism to the Humanities* (Cambridge, MA: Harvard University Press, 1986); and Grendler, *Schooling*.

Young princes and patricians studied Greek and Latin with famous humanist teachers who taught them to see history and the manifest destiny of Italy in new ways.

By the mid-fifteenth century humanist learning far surpassed Petrarch's relatively modest ambitions to converse with Cicero and read an occasional Greek manuscript. Renaissance humanists readily critiqued Petrarch's medieval Latin and wrote in an Italic script that was supposed to simulate how the Romans had written. One particularly apt pupil of Guarino's, Lorenzo Valla (1407–57), became so skilled in the historical evolution of Latin as a language that he wrote a Declamation on the Forged Donation of Constantine (1440), using his philological skills to prove that the evidence for the emperor Constantine's donation of political authority to pope Sylvester I rested on an eighth-century forgery purporting to be a fourthcentury document. The results of the new learning were not always so controversial. The vast majority of humanists devoted themselves to perfecting the rules of grammar, and to historical projects that celebrated the Roman art of historical narrative as a framework within which to interpret the events of Renaissance Italy. Leonardo Bruni, for instance, famously argued that his research proved that Florence was a city of Roman origin rather than one founded by the emperor Charlemagne. The Renaissance city visibly changed with the diffusion of humanist values in society. Albert's neoclassical buildings in Florence and Mantua, for example, reflected his reading of ancient architectural treatises. What lav inside many new palaces also mattered, for many of them now contained private studies. filled with books, coins, paintings, and other artifacts associated with humanist learning.46 Buildings specifically made to hold libraries also appeared, most notably, the Vatican Library conceived by pope Nicholas V in 1451 and inaugurated by Sixtus IV in 1475.⁴⁷

Just as learning embellished the inner spaces of the city, it also played a powerful role in its politics. Coluccio Salutati (1331–1406), the first humanist to become chancellor of the Florentine republic in 1375, in essence the state's chief bureaucrat, expressed well the social and political aspirations of the new learning when he wrote: "Since it is characteristic of man to be taught and the learned are more human than the unlearned, the

⁴⁶ Dora Thornton, *The Scholar in His Study: Ownership and Experience in Renaisance Italy* (New Haven: Yale University Press, 1997); and Findlen, *A Fragmentary Past*. For a discussion of how humanism intersected with art and material culture, see Patricia Fortini Brown, *Venice and Antiquity: The Venetian Sense of the Past* (New Haven: Yale University Press, 1996); and Leonard Barkan, *Unearthing the Past: Archeology and Aesthetics in the Making of Renaissance Culture* (New Haven: Yale University Press, 1999).

⁴⁷ Anthony Grafton, ed., Rome Reborn: The Vatican Library and Renaissance Culture (Washington, D.C.: Library of Congress in Association with the Biblioteca Apostolica Vaticana, 1993).

ancients appropriately referred to learning as humanitas."48 Learning was implicitly a criterion for active membership in the community. The sense of the past that humanists cultivated not only allowed them to reconstruct gloriously Roman histories of their cities, but also provided them with practical skills in great demand in the Italian city-states. The cities of marble, after all, were also cities of paper. Their bureaucracy generated mountains of documents that recorded the daily activities of state – councils, elections, diplomatic relations, and taxes all required the skills of a good humanist education. Sforza's Milan, for example, had so much paperwork that it employed several secretaries in the chancery to organize an archive of state papers and the origins of the Uffizi galleries, which we today think of as a great art museum, lay in the urgent need for more offices (uffizi) to house the Florentine bureaucracy and its papers. Humanists filled an important need in Renaissance politics. They not only espoused the high-minded ideals of Rome with great eloquence, but they also knew how to put their talents to use in writing letters for their rulers and in representing their state abroad, while at home their scholarly skills made them ideal candidates to organize state archives and to examine and authenticate ancient documents of state.

The public uses of learning, however, never entirely precluded alternative interpretations of the importance of a humanist education. "Even though the study of letters promises and offers no reward for women and no dignity," declared the Venetian Cassandra Fedele (1465–1558), "every woman ought to seek and embrace these studies."49 Fedele was one of a handful of women who pursued a humanist education in the fifteenth and sixteenth centuries. Her literacy vastly exceeded the rudimentary knowledge of letters that Margherita Datini acquired from the family notary. Women such as Fedele implicitly challenged the idea that women could not participate fully in all aspects of humanism learning. Fedele publicly demonstrated her eloquence and her mastery of Greek, Latin, history, and rhetoric by giving orations at the University of Padua, to the Venetian people and even before the Venetian doge. Much like the marchesa of Mantua, Isabella d'Este (1474–1539), who was one of many noblewomen who ruled Renaissance states while their husbands were absent from home. Fedele's unusual accomplishments bring into relief the general expectations of learning at this time. More typical was a woman like Alessandra Strozzi who complained to her sons about having to send so many letters to elicit

⁴⁸ Peter Burke, *The European Renaissance: Centres and Peripheries* (Oxford: Blackwell, 1998), p. 29.

⁴⁹ Cassandra Fedele, "Oration in Praise of Letters," in *Her Immaculate Hand: Selected Works by and about the Woman Humanists of Quattrocento Italy*, ed. Margaret L. King and Albert Rabil, Jr. (Binghamton, NY: Medieval and Renaissance Texts and Studies, 1992; 1983), p. 77.

a response from them because "writing seems like hard work." The vast majority of humanist secretaries, laboring to produce and organize reams of paper for the expanding apparatus of state, would have agreed whole-heartedly with her assessment.

At the end of the fifteenth century, the ideal citizen of an Italian citystate was an educated male patrician willing to put his learning to public use, not simply to entertain courtiers (though this was surely what many scholars found themselves doing, writing plays, poems, and histories for public occasions) but to enhance the glory and reputation of his city. He had, of course, excellent taste since in his spare time he read works such as Alberti's On Painting (1434/5) and On the Art of Building (ca. 1450). He could recite poetry spontaneously in the manner of Petrarch and tell a lusty tale worthy of Boccaccio. He was conversant with the leading artists and artisans of the day. He knew the history of Rome and had connected it to the history of whatever city he considered home. One possible candidate for ideal citizenship was the son of an impoverished scholar who so loved Livy's Decades that he made the index for them for a Florentine publisher and sent his seventeen-year-old son to deliver a barrel of good red wine to seal the agreement. 51 This young son not only imbibed the lessons of Livy in his family library, inheriting his father's love of history, but coupled it with an unparalleled thirst for politics. He was the Florentine Niccolò Machiavelli (1469-1527).

4 What Machiavelli Knew About His World

A few months before his death in June 1527, a dispirited Machiavelli told a friend, "I love my native city more than my own soul." Niccolò Machiavelli's love of Florence was not simple hyperbole but the result of extended reflection on the state of the Italian peninsula during the period of the Italian Wars (1494–1530). A brief sketch of what transpired between Petrarch's rediscovery of Rome and Machiavelli's lament for Florence helps us to understand the evolution of Renaissance Italy. Machiavelli had spent his youth in the Florence of Lorenzo de' Medici and had come of age in a city set ablaze by the fiery sermons of the Dominican preacher Girolamo Savonarola (1452–98) who publicly exhorted the Florentines to expell the Medici and rejuvenate their republic. When the Florentines

⁵⁰ Strozzi, Selected Letters, p. 59.

⁵¹ Sebastian de Grazia, *Machiavelli in Hell* (Princeton: Princeton University Press, 1989), p. 6.

⁵² Machiavelli to Francesco Vettori, April 16, 1527, in Maurizio Viroli, *Niccolò's Smile: A Biography of Machiavelli*, trans. Anthony Shugaar (New York: Farrar Straus and Giroux, 2000), p. 254.

answered his call, barring Piero de' Medici's entry to the Palazzo Vecchio in 1494, Savonarola worked with the city leaders to develop a republican government modeled on that of Venice. He expanded membership in the Great Council to approximately 3,000 adult males and laid the groundwork for transforming the position of Standard-Bearer of Justice (*gonfaloniere*) from an elected position that rotated every two months into a position for life, like the office of the doge. Unfortunately Savonarola's aspirations to make Florence the new Jerusalem, purged of errors of faith as well as politics, forced his Florentine allies to choose between Savonarola and the papacy. On May 24, 1498, Savonarola and two associates were burned in Piazza della Signoria⁵³ (Plate 3). Four days later, a twenty-nine year-old Machiavelli learned that he had finally gotten a coveted government post as secretary to the Second Chancery, an office dealing with Florentine territories and foreign relations with other states.

Machiavelli"s entry into Italian politics occurred at a particularly chaotic moment. The death of Lorenzo in 1492 cemented growing discontent with the Medici's oligarchical rule of their city. But the unsettled situation in Florence was simply one of many problems in the Italian peninsula. In the late fifteenth century growing fears about the strength and reach of the Ottoman Empire, beginning with its conquest of Constantinople in 1453, finally reached Italian soil. Popes Nicholas V and Pius II tried to rally the Italian states for a crusade, but found only lackluster support for this initiative, especially on the part of the Venetians whom Pius II accused of preferring a good trading relationship with the Turks to matters of faith. In July 1480 a Turkish fleet of 18,000 men landed near the southern coastal city of Otranto and held it for more than a year until Aragonese troops in the Kingdom of Naples recaptured the city. Such episodes reminded the Italian city-states that their borders were more vulnerable than they had previously thought.

A great deal of this vulnerability, however, came from within. Despite attempts to create a diplomatic accord among the Italian city-states, through the formation of the Italian League in 1455, such agreements were quickly overshadowed by the ambitions of various rulers. The death of Ferrante of Aragon (1458–94) provided a perfect opportunity for the French and Spanish monarchs to claim the Kingdom of Naples as their lawful inheritance, since rule of Naples had passed from the French Angevins to the Spanish Aragonese by force in 1435. The Angevin line being extinct by 1494, it fell to their nearest relatives to claim their inheritance.

⁵³ Donald Weinstein, *Savonarola and Florence: Prophecy and Patriotism in the Renaissance* (Princeton: Princeton University Press, 1970). For an interesting general portrait of the apocalyptic tone of this period, see Ottavia Niccoli, *Prophecy and People in Renaissance Italy*, trans. Lydia G. Cochrane (Princeton: Princeton University Press, 1990).



Plate 3 Anonymous, The Execution of Savonarola (1498 or after). Source: Museo di San Marco, Florence.

This anonymous painting in the convent of San Marco depicts one of the famous events in Florentine history: the burning of the Dominican preacher Girolamo Savonarola (1452-98) and two associates on May 24, 1498 in Piazza della Signoria. After becoming Prior of San Marco in 1491, Savonarola preached publicly of the vanities, excess, and corruption in Medicean Florence at the end of Lorenzo de' Medici's regime and promised his listeners that a new Jerusalem would emerge from the ruins of the city. He famously persuaded Botticelli to give up the pagan themes of his earlier work in favor of a more spiritual kind of painting. The political chaos after Lorenzo's death in 1492 transformed Savonarola from a preacher into a prophet. In an unusual marriage of church and state, he became the guiding force of the Florentine Republic from 1494 until 1497, to such a degree that he even wrote its constitution before his increasingly vitriolic attacks on the papacy and growing divisions within the Florentine government led to his imprisonment, torture, and death. Machiavelli would count him among those "unarmed prophets" who were unable to maintain their power. This painting depicts the most important public piazza, with the medieval symbols of communal government, the Palazzo della Signoria and the Loggia della Mercanzia prominently displayed, and Brunelleschi's cupola to the Duomo also in view.

At the same time, other rulers attempted to use the contest over Naples to their advantage, with unfortunate results. At the end of the fifteenth century, the alliances between various Italian city-states and foreign powers became a pretext for direct involvement in Italian politics. The duke of Milan, Ludovico Sforza, insecure in his own position in Lombardy, invited the French into Italy to destabilize the Aragonese. Charles VIII's army moved swiftly from Milan to Naples, capturing the city by February 1495. Other states allied with the Spanish to expel the French. In 1498 a new French king, Louis XII (1498–1515), began to consider the possibility of deposing his one-time ally Sforza so that he might lay claim to both Lombardy, which he ruled from 1499 until 1513, and Naples. One of the reasons that Leonardo eventually migrated to France was due to his contact with French patrons in Milan in the early sixteenth century.

Around the same time, Ferdinand of Aragon started to contemplate the role that Italy might play in an expanded vision of a Spanish empire that crossed both the Atlantic and the Mediterranean. The years of the Italian Wars were also a crucial period in which explorers and conquistadors such as Columbus and Cortèz made the "Indies" a part of Spain. King Ferdinand's vision proved to be more powerful and better supported within Italy than that of his French rival. By 1504 Spain held Naples, ruling over it until 1707. Ferdinand acquired Milan in 1513. The Spanish temporarily allowed Ludovico Sforza to rule his city again from 1530 until his death in 1535, and subsequently controlled Lombardy until 1713. In 1563, king Philip II established his Supreme Council of Italy to oversee Spain's Italian possession. By 1600 approximately five million of Italy's thirteen million inhabitants were under Spanish rule (see Map 2). 54 The failure of the Italian states to prevent large portions of the peninsula from coming under foreign rule was one of the pressing questions that Machiavelli contemplated as he wrote The Prince.

The papacy, which played an important role in maintaining the Italian League at mid-century, had also become part of the problem. "It is the Church that has kept, and keeps, Italy divided," concluded Machiavelli in *The Discourses on the First Ten Books of Titus Livy* (1516–19).⁵⁵ Machiavelli's well-known fascination with Cesare Borgia (ca. 1476–1507) reflected the emergence of the illegitimate son of pope Alexander VI (1492–1503) as a major force in Italian politics in the late 1490s. The year Machiavelli got his job was the year Borgia renounced his cardinalate in order to pursue his ambition to rule Romagna. Machiavelli did not meet Borgia until 1502, when he had conquered much of central Italy. But he had heard of his

⁵⁴ Gregory Hanlon, Early Modem Italy. 1550–1800 (New York: St. Martin's Press, 2000), p. 74.

⁵⁵ Machiavellli, The Discourses, p. 145.



Map 2 Italy ca. 1600.

talent and his aspirations. When he came to know Borgia, he admired his skill in creating and maintaining a state by force.

The Venetians also faltered at this crucial moment in Italy's history. Having largely used their wealth and geographic isolation to insulate themselves from the problems facing many other Italian states, they found themselves with few allies. On May 14, 1509, French troops defeated the Venetians at Agnadello. Most of their mainland territories surrendered to their enemies (king Francis I, emperor Maximilian I, pope Julius II, and the duke of Ferrara) within a few weeks. While the Venetians eventually regained their possessions by 1517, their political standing within Italy was considerably weakened by the $1530s.^{56}$ Machiavelli would later observe that the lack of a strong army and their historic unwillingness to work with the papacy had left Venice vulnerable to such an attack.

It was an auspicious moment, in other words, to be well-versed in politics. Machiavelli, the offspring of an illegitimate member of a respectable Florentine family, felt that his position as secretary offered him unique opportunities to rub shoulders with the leading figures of his time – Cesare Borgia, Leonardo da Vinci, Isabella d'Este were but a few of the people he encountered in his travels on behalf of the troubled Florentine Republic. Years later, after the fall of the Florentine Republic in 1512 which forced Machiavelli into a life of leisure because his services were not wanted by the new Medici government, he reflected on the possibilities that lay before him. He was no merchant, he confessed. He had little land, few books, good Latin, and a gift with words. What should such a person do? He concluded: "Not knowing how to reason about the art of silk or about the art of wool, or about profits and losses, it is necessary that I reason about the state."57 Thinking about politics might be a poor substitute for being in the thick of things, but he hoped that the fruits of his reflection might earn him a new office in Medicean Florence and assist Florence's rulers in avoiding the mistakes of the past.

There is no doubt that Machiavelli felt somewhat disaffected from his world. He had been forced out of government, one of few to lose their posts with the return of the Medici, and spent most of his days at his country villa in Sant'Andrea in Percussina, reading, writing, and thinking. In the spring of 1513, as he completed *The Prince* and filled his leisure hours in exile reading the poetry of Dante, Petrarch, and Ovid when he felt playful, and

⁵⁶ Robert Finlay, "Fabius Maximus in Venice: Doge Andrea Gritti, the War of Cambrai, and the Rise of Habsburg Hegemony, 1509–1530," *Renaissance Quarterly* 53 (2000): 988–1031. These events are often described as the War of the League of Cambrai (1509–17), referring to the League made by Venice's enemies to defeat and partition the Venetian Republic.

⁵⁷ De Grazia, *Machiavelli in Hell.* p. 132. This passage comes from a letter to Francesco Vettori, Florentine ambassador to the pope, written April 9, 1513.

Cicero and Livy when he wished to contemplate weighty affairs, he had only recently returned from months of imprisonment in the Bargello where he had been tortured on suspicion of conspiring against the Medici who had returned to Florence precisely because they allied themselves with the Spanish and now sat on the papal throne (Leo X, 1513–21 and Clement VII, 1523–34). In the coming years, Machiavelli's literary production mitigated his sense of disaffection. Works such as his *Discourses* and *The Art of War* (1521) reflected conversations he had with young men in the Orti Oricellari, the gardens of Bernardo Rucellai's palace where leading citizens gathered to discuss politics and culture. They reflected Machiavelli's growing ambition to get his Renaissance readers to see how history might offer a vital resources for understanding the modern problems of state, not to mention the peculiarities of Italian politics that looked increasingly dated in a world subject to the growing power of France and Spain.

Machiavelli's reemergence in society also manifested itself in his plays, which only tangentially referred to politics and instead offered sharp observations about life in a Renaissance city. The city of Florence after 1512 seemed curiously empty of its leading artists and writers because so many, like Michelangelo and Leonardo, had immigrated to Rome with the election of Leo X in 1513, making Machiavelli's occasional presence in the city even more welcome. Audiences laughed heartily at The Mandrake (1518), a tale of lust and deceit that subtly dissected the Florentine obsession with wealth and lineage, presenting the city as a world teeming with middlemen who knew how to close a deal. "This is your Florence," he declared in the Prologue.⁵⁸ The Florentines also appreciated his poetry. though not everyone would have agreed with Machiavelli's own opinion that his verses were so good that he deserved a place in Ludovico Ariosto's list of great poets in Orlando Furioso (1516).⁵⁹ A great prose writer surely, however, since all of Machiavelli's works were written in Tuscan rather than Latin, reflecting the growing trend towards using the vernacular in the sixteenth century.

Machiavelli's *Florentine Histories* (1520–5) represented the culmination of his efforts to present his version of the past. It was also the only work that Machiavelli wrote under the patronage of the Medici. While he dedicated *The Prince* to Leo X's nephew, Lorenzo de' Medici (1492–1519), it was cardinal Giulio de' Medici, appointed by his cousin Leo X to rule Florence after Lorenzo's death, who specifically commissioned Machiavelli's history of Florence. Giulio, the future Clement VII, seems to have been the only Medici who decided to put Machiavelli's talents to use. He asked his advice on how to rule the city in 1519. When Clement VII became pope in 1523,

⁵⁸ Niccolò Machiavelli, *Mandragola*, ed. and trans. Mera J. Flaumenhaft (Prospect Heights, IL: Waveland Press, 1981), p. 9.

⁵⁹ De Grazia, Machiavelli in Hell, p. 48.

Machiavelli must have felt that times were auspicious for his return to public service. And skilled men were urgently needed as Florence prepared itself for the renewal of war. In spring 1526, shortly before Machiavelli's death, he was sent out to inspect public buildings and fortifications – a small job perhaps but well suited to the author of *The Art of War*.

In the 1520s, as the artist Benevenuto Cellini (1500-71) observed, "The whole world was now at war."60 Machiavelli did not live to see the emperor Charles V's German and Spanish troops sack the papal city on 6 May 1527. an event which left its mark not only in the decimation of the populace but also in more whimsical ways such as the graffiti of a German soldier who carved the name "Luther" into one of Raphael's frescoes in the Vatican. 61 Clement VII was held virtual prisoner in Castel Sant'Angelo and subsequently fled Rome for refuge in Orvieto until 1528. Subsequent agreements between the pope and the emperor made it clear who had the upper hand in Italy. On February 24, 1530, Clement VII crowned Charles V emperor in Bologna. Imperial troops were then in the process of besieging the city of Florence, which had once again rejected the Medici with the establishment of the Second Florentine Republic in 1527. Spanish troops entered a starving and beleaguered city in August 1530. The following year, Charles V forced the Florentines to accept the hereditary rule of the Medici and forced the Medici to accept his sovereign authority. On May 1, 1532, the old signoria that had shaped communal rule in Florence since the late thirteenth century, was abolished. The Medici proclaimed their new status by making the Palazzo della Signoria their residence as well as the center of the new state. They were now truly princes, reflecting a growing preference for this form of governance in sixteenth-century Italy, despite the continuation of republics such as Venice and Lucca.

5 Ending the Renaissance

The end of republican rule in Florence certainly did not signal the end of the Renaissance, because there were too many different aspects to the Italian Renaissance for the conclusion of a single episode to determine the outcome. Instead, the transformation of Florence from a republic to the capital city of the Grand Duchy of Tuscany might well be the logical culmination of certain tendencies already inherent in Italian society throughout fifteenth century such as the consolidation of states under hereditary rulers. Comparing Lorenzo "Il Magnifico" with his grandson Lorenzo de' Medici in 1516, Lodovico Alamanni noted that he could not

⁶⁰ Benvenuto Cellini, Autobiography, trans. George Bull (London: Penguin, 1956), p. 69.

⁶¹ André Chastel, *The Sack of Rome, 1527.* trans. Beth Archer (Princeton: Princeton University Press, 1983), p. 92. See also Kenneth Gouwens, *Remembering the Renaissance: Humanist Narratives of the Sack of Rome* (Leiden: Brill, 1998).

mingle informally with citizens as his illustrious grandfather had done "because being in such a grand position, it would be improper for him to act like a private citizen." If Florentines initially were "unfamiliar with the ways of courts," as Alamanni observed, they soon learned to treat the Medici as rulers rather than first citizens.⁶²

Politics was a different sort of preoccupation by the end of the sixteenth century, when the former Jesuit Giovanni Botero (1544–1617) composed The Reason of State (1589), based on his experiences in Spanish Milan, the duchy of Savoy, and papal Rome. Botero took for granted some of the issues that Machiavelli had begun to explore such as the image of Italy shaped by an accord between the Spanish and the pope. He dismissed Machiavelli's perceptive comments about the attractions of rule by ability and force (what Machiavelli called *virtù*), preferring instead to underscore the importance of the right to rule and the effects of just rule. He downplayed the idea that the papacy had played an unfortunate role in Italy's troubles, as the pinnacle of princely ambition in the Renaissance game of politics, by underscoring the union between politics and faith as the basis for legitimate rule. Botero. in other words, reflected the results of Italian politics after the Spanish conquest of Italy and the difficult period of the Reformation, when the emergence of Protestantism redrew the political and religious map of Europe and led Catholics to define more precisely what it mean to be a good Catholic in the decrees of the Council of Trent (1545–63). He wrote in light of a renewed and expanded papacy and a flourishing of Catholicism not just in Italy and Catholic Europe, but throughout the world in the era of the Jesuit and Franciscan missions that Christianized parts of Asia and the Americas. 63 He wrote of politics in light of Machiavelli and in recognition of the transformation of Italy in his own century.

The first decades of the sixteenth century were a moment of profound nostalgia. Machiavelli's close friend Francesco Guicciardini (1483–1540), who had successfully worked both for the Florentine republic and for the Medici, found himself unexpectedly out of favor in 1537. His spent the last years of his life composing *The History of Italy* (1540) so that posterity would understand how "the misfortunes of Italy" (*le calamità d'Italia*) between 1492 and 1530 had occurred. The underlying premise of Guicciardini's tragic narrative was the idea of a golden age, culminating in the rule of Lorenzo de' Medici. Italy in 1490, he informed his readers, had seemed equal to the best moments that history recorded of the Roman empire. Such nostalgia led later generations of scholars, reading

⁶² Kohl and Smith, *Major Problems*, p. 443. This was the same Lorenzo to whom Machiavelli dedicated *The Prince*.

⁶³ John M. Headley, "Geography and Empire in the Late Renaissance: Botero's Assignment, Western Universalism, and the Civilizing Process," *Renaissance Quarterly* 53 (2000): 1119–55.

Guicciardini's words, to imagine the unraveling of Italian politics in these decades as the end of the Renaissance.

Less than ten years before Guicciardini began his history, a Mantuan aristocrat who spent the majority of his life in the service of the dukes of Urbino put the finishing touches on a portrait of Renaissance court life under the rule of Guidobaldo da Montefeltro (1472-1508) and Elisabetta Gonzaga (1471–1526). Baldassare Castiglione (1478–1529) wrote The Book of the Courtier (1528) as a description of the qualities of a perfect courtier that emerged in four evenings of imaginary conversation at the court of Urbino. In discussions monitored by Elisabetta Gonzaga and her noblewomen, male courtiers defined the best attributes of male and female courtiers and described how their skillful display of sprezzatura – the ability to make difficult things look easy - would win them praise and honor (Plate 4). In the same period, the satirist Pietro Aretino (1492–1556) dissected the values that Castiglione celebrated by offering up bawdy portraits of Roman courtesans and prostitutes who perpetuated the illusions of their noble clientele that paying for their attentions was not all about sex, simply because they could recite Petrarchan verse. 64 Several decades later, the Florentine Giovanni della Casa (1503–56), who spent his youth at the papal court and later served as archbishop of Benevento and papal nuncio (ambassador) in Venice, composed a short response to Castligione's Book of the Courtier in the form of his Galateo (1558). Delia Casa astutely observed that the first lesson of courtiership concerned the regulation of the body, its appetites, and its embarrassing intrusions into the heady world of politics and culture. No conversation about the virtues of the ancients would matter, he intoned, if one did not know how to eat gracefully, defecate discreetly, and avoiding showing one's companions snotty and crumpled handkerchiefs.65

Like Guicciardini, Castiglione wrote of Urbino as if it were in the past. True, the principal inhabitants of this version of Urbino were dead by the time he finished the book, though a number of its key protagonists – pope Leo X's secretary Pietro Bembo (1470–1547), a Venetian who argued that Tuscan should be the basis for good Italian prose regardless of one's region of origin and became a cardinal in 1539, and the poet Bernardo Accolti (1458–1535) – still lived. Many had read drafts of the book during the late 1510s, participating in the collective burnishing of the mirror of their

⁶⁴ Paula Findlen, "Humanism, Politics, and Pornography in Renaissance Italy," in *The Invention of Pornography*, ed. Lynn Hunt (New York: Zone, 1993), pp. 49–108; and Bette Talvacchia, *Taking Positions: On the Erotic in Renaissance Culture* (Princeton, NJ: Princeton University Press, 1999).

⁶⁵ Antonio Santosuosso, "Giovanni Della Casa and his *Galateo*: On Life and Success in the Late Italian Renaissance," *Renaissance and Reformation* 11 (1975): 1–13.



Plate 4 Agnolo Bronzino, *Portrait of Ugolino Martelli* (ca. 1535–8). *Source*: Staatliche Museen, Berlin.

Agnolo di Cosimo di Mariano Tori, known as Bronzino (1503–72) was one of the leading court painters in Florence during the sixteenth century and also a poet. Bronzino first came to the attention of the Medici when he arrived in Florence to work on decorations for the 1539 marriage of duke Cosimo I and Eleonora of Toledo, daughter of the viceroy of Naples. He soon became the leading portrait painter of the Medici court, producing numerous images of idealized courtiers as well as the official portraits of the Medici family as they established themselves as the hereditary grand dukes of Tuscany. Bronzino's portrait of Ugolino Martelli was completed after he spent two years (1530–32) in Urbino, the setting for Baldassare Castiglione's *Book of the Courtier* (1528). His painting captures visually many of the attributes of a good courtier described by Castiglione. Martelli was a young Florentine scholar who is presented as the embodiment of the virtues of learning and art. His hand rests on the ninth book of Homer's *Iliad*. Behind him is a miniature reproduction of Michelangelo's *David* (1504). His black garments reflect the new taste for dressing in emulation of Spanish courtiers.

society. But the activities that they described did not disappear in the next century. The cultural pursuits such as poetry, music, and painting that Castiglione celebrated formed the basis for court life in the sixteenth and seventeenth centuries. The wide circulation of his book – some fifty-eight editions in Italy alone in the sixteenth century – ensured that many literate residents of the Italian cities also strove to emulate the virtues of the courtier, reciting poems, discussing paintings, joking with Boccaccio, and carefully managing their appearance to make a good impression on others. Such popularity suggests how humanist values enjoyed a significantly broader audience in late Renaissance Italy than they had during the inception of this intellectual movement. A half-century later, when the poet Torquato Tasso (1544–95) composed *Malpiglio, or On the Court* (1585), he could talk about a young man who "has read Castiglione's *Courtier* and has also memorized it."

By the time Tasso wrote of a young courtier who knew Castiglione better than Cicero, one important change had taken place that directly affected the reading of many classics of Italian Renaissance literature. The reformed Catholicism that emerged from the Council of Trent led to the establishment of the Index of Prohibited Books in 1557. While primarily concerned with banning writings by heretical authors, the Index also identified authors whose writings were morally questionable. The works of Machiavelli and Aretino were banned outright. Lorenzo Valla's critical comments on the church's dubious temporal authority also ended up on the Index. Boccaccio's Decameron and Castiglione's Book of the Courtier were carefully expurgated to remove offending passages about the sex lives of clerics, the political ambitions of popes, and of course pagan concepts such as Machiavelli's idea of fortune (fortuna) that did not give due authority to God as the agent who shaped the world. As a result, late sixteenth-century readers saw these works in light of the potential conflict their ideas posed with the current teachings of the Catholic Church. But they did not cease to read the books of the Renaissance. Putting works on the Index virtually ensured them an audience. Since the majority of Italian books were published in Venice, a city that openly defied the pope on more than one occasion, no one could really complain that it was hard to get a copy of *The Prince*. 67

⁶⁶ Torquato Tasso, *Tasso's Dialogues*, ed. and trans. Garnes Lord and Dain A. Trafton (Berkeley: University of California Press, 1982), p. 155. Peter Burke, *The Fortunes of the Courtier* (University Park, PA: Pennsylvania State University Press, 1996; 1995).

⁶⁷ Paul Grendler, *The Roman Inquisition and the Venetian Press*, 1540–1605 (Princeton: Princeton University Press, 1977). The classic study of heretical reading in this period remains Carlo Ginzburg, *The Cheese and the Worms: The Cosmos of a Sixteenth-Century Miller*, trans. John and Anne Tedeschi (Baltimore: Johns Hopkins University Press, 1980). Heresy in general in this period is well discussed in John Martin, *Venice's Hidden Enemies: Italian Heretics in a Renaissance City* (Berkeley: University of California Press, 1993).

While the post-Tridentine Catholic Church may have taken a dim view of the excessive paganism of some Renaissance humanists, whose love affair with antiquity was not always easily reconciled with Christianity, they assisted the institutionalization of Renaissance learning in other ways. It was one of the new religious orders, the Society of Jesus (founded in 1540). that made humanist learning the centerpiece of their educational system. building on the tradition of Christian humanism that had already been a strong part of Roman intellectual life since the late fourteenth century.⁶⁸ The first Jesuit college opened in Messina in 1548. By 1600 there were forty-nine Jesuit colleges in Italy, and that number more than doubled in the seventeenth century. 69 The Jesuits educated the sons of many prosperous Italian families as well as their own novices. Their colleges formalized the humanist study of ancient languages and rhetoric in ways that the fifteenth-century academies and universities largely had been unable to do. In this respect, the Catholic Church fully embraced the idea of humanism as the foundation for Catholic education.

The arts as well as learning continued to thrive in late Renaissance Italy. When Giorgio Vasari wrote his *Lives of the Artists*, he did not do this because the age of great painting and sculpture had ended but because he considered himself to be living at its pinnacle. Michelangelo was still alive and working in Rome when Vasari published his first edition of 1550. He had only been dead four years when the second edition appeared in 1568. Vasari's narratives of the growth and perfection of the arts in Italy diverged from Guicciardini's dismal account of the tragedy of Italy in many important ways. The political breaks that Guicciardini saw in 1492, 1494, and 1527 hardly mattered to Vasari, because art had not disappeared as a result of the death of Lorenzo de' Medici, the invasion of the French king Charles VIII, or the Sack of Rome. Ouite the opposite since the Rome of the Farnese pope, Paul III (1534–49), enjoyed the presence of Michelangelo and many other talented painters and sculptors, while the artistic programs in many other cities - Mantua, Florence, and Venice, to name a few - were flourishing. In many respects, the true heyday of Renaissance Rome lay not in the fifteenth century but at the end of the sixteenth century when many of the haphazard building projects of a succession of popes and cardinals made way for a more systematic approach to urban planning.

Such activities easily contradict the prevalent view that the political and religious disasters of the early sixteenth century effectively brought the Renaissance to a close. Both the sixteenth-century courts and church

⁶⁸ John W. O' Malley, *Praise and Blame in Renaissance Rome* (Durham, NC: Duke University Press, 1979).

⁶⁹ Grendler, Schooling, p. 371; and John W. O'Malley, The First Jesuits (Cambridge, MA: Harvard University Press, 1993).

increased their support for many of the activities that we associate with the idea of the Renaissance. If there is one major change that we can point to. it is perhaps the economy. Renaissance Italy had emerged as a result of the commercial vitality of the Mediterranean and of the unique goods and services that various Italian city-states offered to the rest of Europe. Despite the warfare and changes of government in the early sixteenth century, much of this economy remained in place for the next century. A wide array of Italian cities continued to offer unique goods and services. Venice continued to be an important port of call, while the Tuscan city of Livorno (Leghorn) emerged as one of the new ports connecting the Atlantic and Mediterranean economies. In the course of the seventeenth century, however, Italy's overall economic prosperity seems to have eroded. ⁷⁰ In part, this was because of new international upheavals such as the Thirty Years' War (1618-48), but it also reflected the emergence of the Atlantic world economy which made cities such as Seville, London, and Amsterdam strong competitors with the Italian ports, and produced new and cheaper goods that competed successfully with many of the high quality products that Italian artisans continued to produce. Some have argued that the strength of the guild system in many Italian cities was ultimately a liability in relation to newer economic initiatives that favored a larger scale production over more careful regulation of different industries.

Yet even the transformation of the Italian economy was too slow and uneven for us to conclude that it triggered the end of the Renaissance. So we are left with a puzzling question: how exactly should we describe the transformation of Renaissance Italy into something different, something beyond the Renaissance, in the course of the sixteenth and seventeenth centuries? Most recently, William Bouwsma has suggested that we see the period from 1550 to 1640 as a "waning of the Renaissance" – a gradual cessation of the values, practices, and beliefs that characterized the idea of a rebirth of society. Such an image holds a great deal of appeal because it presents the Renaissance as a general cultural ideal that manifested itself in many different forms before fading into the background to allow room for new political ideas, religious practices, and intellectual initiatives.

⁷⁰ Carlo Cipolla, "The Economic Decline of Italy," *Economic History Review* ser. 2, 5 (1952–53): 176–87; and Judith Brown, "Prosperity or Hard Times in Renaissance Italy?" *Renaissance Quarterly* 42 (1989): 761–80. For a more general account, see Domenico Sella, *Italy in the Seventeenth Century* (New York: Longman, 1997), pp. 19–49.

⁷¹ William J. Bouwsma, *The Waning of the Renaissance* 1550–1640 (New Haven: Yale University Press, 2000). For other discussions of the end of the Italian Renaissance, see especially Eric Cochrane, ed.. *The Late Italian Renaissance* (New York: Harper Torchbooks, 1970); idem, *Italy* 1530–1630, ed. Julius Kirshner (New York: Longman, 1988); Peter Burke, *The European Renaissance: Centres and Peripheries* (Oxford: Blackwell, 1988); and Hanlon, *Early Modern Italy*.

The Italian Renaissance did not simply end, in some definitive sense. If this were true, we would not be studying the humanities in our own universities, nor would we place such a premium on historical thought as a fundamental aspect of the human experience. We continue to find Petrarch's reflections on the problems of living in one's own time, Machiavelli's astute comments on how to maintain power, and Castiglione's arguments about the virtues of self-presentation relevant to our own concerns. Italy ceased to be the epicenter of western Europe by the late seventeenth and eighteenth centuries, when France and England came to dominate discussions of politics, economy, and culture, and when their trans-Atlantic empires became as large and influential as the one the Spanish created in the sixteenth century. But Italy did not cease to capture the imagination of foreigners, who traveled to Italy in greater numbers after the Renaissance than they had done in the era when it seemed to be the center of the world.

Foreigners increasingly came to Italy on the Grand Tour not only to see its Roman antiquities but also to admire the tangible accomplishments of Italian Renaissance society. In 1620, for example, Michelangelo's descendants opened the Casa Michelangelo so that art connoisseurs could see examples of the work that Vasari celebrated.⁷² By the late seventeenth century the great Florentine astronomer Galileo Galilei's telescope was on display in the grand ducal galleries of the Uffizi next to images such as Raphael's Pope Leo X with Two Cardinals (1517). Guidebooks took visitors through the piazzas, churches, and palaces of Italy, offering learned commentary on what was worth seeing of Italy's past. Vasari's rinascita became more than simply a word describing the process by which a new vision of the arts emerged between the fourteenth and sixteenth centuries. It became a historical label for an age that was increasingly removed from the present. Time allowed the idea of the Renaissance to take shape and give coherence to the diversity of activities that defined Italian society in this period. In this respect, the Italian Renaissance is a direct product of our own engagement with Petrarch. He polemically labeled the period since the fall of the Roman empire to his own time the "Dark Ages." We have responded by agreeing with Vasari that the subsequent age should be called the "Renaissance." As the argument for the Renaissance as the birth of the modern world has receded, this label itself has been subject to inspection. 73 But the fascination with the society that produced this word remains.

⁷² Burke, European Renaissance, p. 200.

⁷³ Randolph Starn, "Who's Afraid of the Renaissance?" In *The Past and Future of Medieval Studies*, ed. John van Engen (Notre Dame: University of Notre Dame Press, 1994), pp. 129–47; and Anthony Molho, "The Italian Renaissance, Made in the USA," in *Imagined Histories: American Historians Interpret the Past*, ed. Anthony Molho and Gordon S. Wood (Princeton: Princeton University Press, 1998), pp. 263–94.